Author’s response to reviews

Title: A mouse model of sitosterolemia; absence of Abcg8/sterolin-2 results in failure to secrete biliary cholesterol

Authors:

Dr Eric L Klett (klettel@musc.edu)
Kangmo Lu (luk@musc.edu)
Astrid Kosters (A.Kosters@amc.uva.nl)
Edwin Vink (E.Vink@amc.uva.nl)
Mi-Hye Lee (leemih@musc.edu)
Michael Altenburg (mikelly@email.unc.edu)
Sarah Shefer (salenge@umdnj.edu)
Ashok K Batta (ashkbatta@hotmail.com)
Dr Hongwei Yu (yuh@musc.edu)
Jianliang Chen (chenjl@musc.edu)
Richard Klein (kleinrl@musc.edu)
Norbert Looije (N.Looije@amc.uva.nl)
Ronald Oude-Elferink (r.p.oude-elferink@amc.uva.nl)
Albert K Groen (a.k.groen@amc.uva.nl)
Nobuyo Maeda (nobuyo.maeda@pathology.unc.edu)
Gerald Salen (salenge@umdnj.edu)
Dr Shailendra B Patel (patelsb@musc.edu)

Version: 2 Date: 17 Nov 2003

Figures 2 and 9 have an error in formatting. The units for the y-axes should read ug/gram liver. Somehow the u was left off the originally submitted figures.